Author's response to reviews

Title: Diagnosis and follow-up of treatment of latent tuberculosis; the utility of the QuantiFERON-TB Gold In-tube assay in outpatients from a tuberculosis low-endemic country

Authors:

Anne M Dyrhol-Riise (adri@helse-bergen.no)
Gerd Gran (gerd.gran@helse-bergen.no)
Tore Wenzel-Larsen (tore.wenzel-larsen@helse-bergen.no)
Bjørn Blomberg (bjorn.blomberg@med.uib.no)
Christel Gill Haanshuus (christel.gill.haanshuus@helse-bergen.no)
Odd Mørkve (odd.morkve@cih.uib.no)

Version: 3 Date: 14 February 2010

Author's response to reviews:

Dear Editor

Please, se attached our revised manuscript. We appreciate that the manuscript is accepted for publication in BMC Infectious Diseases.

As suggested by the editor we have updated figure 4 and amended both figure 1 and figure 2 and we hope they are improved.

We have further tried to improve the manuscript once more in response to the comments made by the reviewers and gone through the formatting of the manuscript. We hope that the manuscript now is acceptable for publication.

Sincerely

Anne Ma Dyrhol-Riise
Corresponding author